\newcommand{\Comment}[1]{}

\chapter{ $B$O$8$a$K(B \Comment{(Introduction)} }
\label{intro}

$BK\>O$G$O!"K\O@J8$N35MW$r=R$Y$k!#(B

\vspace{1em}

% $B6qBNE*$K(B!

% DKLIC $B$H$O2?$+(B

$BIa5Z$7$F$$$k<jB3$-7?8@8l$G$O!"(B
$B%W%m%;%972$NJ,;6!"F14|!"DL?.%W%m%H%3%k$N<BAu$H$$$C$?(B
$BJ,;6%W%m%0%i%_%s%0$KIT2D7g$J=hM}$N5-=R$,Hs>o$KLLE]$G$"$k!#(B

$BK\8&5f$G$O!"(B
$BF0E*%W%m%;%9@8@.!"<+F0F14|!"HsF14|%a%C%;!<%8DL?.(B
$B$r@k8@E*$K5-=R$9$k(B KL1 $B8@8l$K4p$E$-!"(B
$B4{B8$N(B KLIC $B=hM}7O$r3HD%$7$F(B
$B@k8@7?%=%1%C%HDL?.$N5!G=$rDs6!$9$k(B DKLIC $B=hM}7O$r<BAu$7$?!#(B
%
$B$3$N$3$H$K$h$C$F!"D>4QE*$G$J$+$C$?J,;6=hM}$N5-=R$r(B
$B@k8@E*$K9T$($k$h$&$K$7$?!#(B

$BJ,;6O@M}JQ?t$ND>7k$dGQ4~$r8!=P$9$k$3$H$K$h$C$F!"(B
$B4{B8$N(B DKLIC $B=hM}7O$G$OIT2DG=$G$"$C$?(B
$BFs=EAw?.$dDL?.$NCf7Q$rGS=|$7!"(B
$BJ,;6O@M}JQ?t$N(B ID $B$rAa4|$K:FMxMQ$9$k$3$H$r2DG=$H$7$?!#(B

$B$^$?!"(BKL1 $B8@8l$KC_@Q$5$l$?9bEY$J@EE*2r@O$N5;=Q$r(B
$BJ,;6%W%m%0%i%_%s%0$KE,MQ$9$k$3$H$r2DG=$H$7$?!#(B
%
$B$=$l$@$1$G$J$/!"4{B8$N@EE*2r@O$N5;=Q$G$"$k(B
$B7?!&%b!<%I!&@~7A@-BN7O$KL@3N$K$O8=$l$J$$(B
$BDL?.O)$N3NN)!JJ,;6O@M}JQ?t$N(B open$B!K$H$$$&35G0$rL@3N$K<($7$?!#(B
%
$BJ,;6%W%m%0%i%_%s%0$K$*$$$F$O!"(B
$BAG@-$NCN$l$J$$DL?.%G!<%?$d%b%P%$%k%3!<%I$KBP$7$F(B
$B@EE*$K8!>Z$r9T$($k$3$H$OHs>o$K=EMW$G$"$k!#(B

$B$^$?!"@EE*0UL#O@$rJQ$($:$KMM!9$JDL?.<jK!$r;n$_$k$3$H$O(B
$B:#8e$N%M%C%H%o!<%/$NIi2Y$r7Z$/$9$k8&5f$KLrN)$D$G$"$m$&!#(B
%
$BDL?.<jK!$K$h$k%M%C%H%o!<%/$NIi2Y$NJQ2=$rDjNLHf3S$9$k$3$H$O!"(B
$BK\8&5f$K$*$$$F$b:#8e$N2]Bj$G$"$k!#(B


\section{ $BK\O@J8$N9=@.(B }

$BK\>O$G$O!"K\O@J8$N35MW$r=R$Y$?!#(B

$BBh(B\ref{dist}$B>O$G$O!"J,;6%W%m%0%i%_%s%0$N8=>u$HLdBjE@$r=R$Y$k!#(B

$BBh(B\ref{KLIC}$B>O$G$O!"(BKL1 $B8@8l$N4pK\$H4{B8$N(B
KL1 $B8@8l=hM}7O$K$D$$$F2r@b$9$k!#(B

$BBh(B\ref{analysis}$B>O$G$O!"(B
$BK\8&5f$,A0Ds$H$9$kJ,;64D6-$HK\8&5f$NL\I8$r@_Dj$9$k!#(B

$BBh(B\ref{DKLICP}$B>O$G$O!"(BDKLIC $B$,;HMQ$9$kDL?.%W%m%H%3%k$r7hDj$9$k!#(B

$BBh(B\ref{design}$B>O$G$O!"(BDKLIC $B$N;EMM$H@_7W$r=R$Y$k!#(B

$BBh(B\ref{impl}$B>O$G$O!"(BDKLIC $B$r<BAu$9$k!#(B

$BBh(B\ref{eval}$B>O$G!"(BDKLIC $B$rI>2A$7!"9M;!$r2C$($k!#(B

$BBh(B\ref{conc}$B>O$G!"K\O@J8$r$^$H$a$k!#(B
